Title: Trends in antibiotic use among outpatients in New Delhi, India

Authors:

Anita Kotwani (anitakotwani@gmail.com)
Kathleen Holloway (hollowayk@searo.who.int)

Version: 5 Date: 16 April 2011

Author’s response to reviews: see over
April 16, 2011

Dear Editor,

I am submitting the revised manuscript. R4 (MS: 1694694726487911), entitled, “Trends in antibiotic use among outpatients in New Delhi, India”.

I have revised the manuscript on the lines suggested by the editorial office.

Titles and Legends for the figures are after the references in the MS. Figures are numbered 1-6 with the title. Figure numbers are changed at appropriate place in the MS.

All the highlighting and tracking have been removed.

Please remove the visible vertical lines from tables at your end.

Many Thanks.

Best Regards,

Anita Kotwani